Author's response to reviews

Title: MicroRNA-1269 Promotes Proliferation in Human Hepatocellular Carcinoma via Downregulation of FOXO1

Authors:

XueWei Yang (yangxw0908@163.com)
Guan-Zhu Shen (lawysun@163.com)
LiangQi Cao (clq0829@163.com)
XiaoFeng Jiang (jiangxiaofeng@163.com)
HePing Peng (penghp@163.com)
Gang Shen (13066336440@163.com)
De Chen (drchende@163.com)
Ping Xue (drxueping@medmail.com.cn)

Version: 5 Date: 14 November 2014

Author's response to reviews: see over
Author's response to editor

Title: MicroRNA-1269 Promotes Proliferation in Human Hepatocellular Carcinoma via Downregulation of FOXO1

Authors:
Xue-Wei Yang (Email:yangxw0908@163.com)
Guan-Zhu Shen (Email:lawysun@163.com)
Liang-Qi Cao (Email:clq0829@163.com)
Xiao-Feng Jiang (Email:jiangxiaofeng008@163.com)
He-Ping Peng (Email:penghp@163.com)
Gang Shen (Email: 13066336440@163.com)
De Chen (Email:drchende@163.com)
Ping Xue (Email:drxueping@medmail.com.cn)

Version: 3 Date: 14 November 2014
Author's response to editor: see over
Nov 14th, 2014

Dr. Kazushi Sugimoto
BioMed Central
Floor 6, 236 Gray's Inn Road
London
WC1X 8HB
United Kingdom

Dear Dr. Sugimoto,

Thank you for your decision letter dated Nov 13th, 2014, regarding our manuscript MS 1182039190135524, entitled “MicroRNA-1269 promotes proliferation in human hepatocellular carcinoma via post-transcriptional downregulation of FOXO1” by Xue-Wei Yang et al. We thank for the editor’s comment. According to the editor’s comment, we have revised the Methods section of our manuscript to include the name of the ethics committee, “Sun Yat-sen University Cancer Center Institutional Review Board”, to approve our study.

We are now re-submitting a revised manuscript and we hope that you will find that the current version of our manuscript is suitable for publication in BMC Cancer. Your supportive consideration is greatly appreciated.

Please do not hesitate to contact us if there is anything we could do to help improve the manuscript.

Sincerely yours,

Ping Xue MD.
Department of Hepatobiliary Surgery
The Second Affiliated Hospital of Guangzhou Medical University
Guangzhou, 510260, China.
Email: drxueping@medmail.com.cn;
Phone: 86-20-34152282
Fax: 86-20-34153080.